Author's response to reviews

Title: Cluster randomized trial in smoking cessation with intensive advice in diabetic patients in primary care. ITADI Study

Authors:

Lydia Roig (31540lrr@comb.cat)
Santiago Perez (estudi.itadi@gmail.com)
Gemma Prieto (gpljenni@gmail.com)
Carlos Martin (carlos.martin@uab.cat)
Mamta Advani (mamtaadvani@gmail.com)
Angelina Armengol (anarmengol@gmail.com)
Pilar Roura (roura@cusido.com)
Josep Maria Manresa (jmanresa@idiapigol.info)
Elena Briones (elena.m.briones@gmail.com)
ITADI Study Group (estudi.itadi@gmail.com)

Version: 8 Date: 2 February 2010

Author's response to reviews: see over
February 2, 2010

Editorial office: Bio Med Central Public Health

Dear Editor,

Re: Revisions of protocol in BMC Public Health

Please find attached our protocol entitled “Cluster randomized trial in smoking cessation with intensive advice in diabetic patients in primary care. ITADI Study”, which includes the revisions made by the BMC editors. We have included the following edits:

1. In the acknowledgements section, we corrected a typographical error, changing “The authors also thanks all of the associated…” to “The authors also thank all of the associated…”

Yours sincerely,

Santiago Pérez, M.D.